## Terlipressin for hepatorenal syndrome (Review)

Gluud LL, Christensen K, Christensen E, Krag A



This is a reprint of a Cochrane review, prepared and maintained by The Cochrane Collaboration and published in *The Cochrane Library* 2012, Issue 9

http://www.thecochranelibrary.com

## WILEY

## TABLE OF CONTENTS

HEADER	1
ABSTRACT	1
PLAIN LANGUAGE SUMMARY	2
BACKGROUND	2
OBJECTIVES	3
METHODS	3
RESULTS	4
Figure 1	5
Figure 2	7
Figure 3	9
DISCUSSION	9
AUTHORS' CONCLUSIONS	11
ACKNOWLEDGEMENTS	11
REFERENCES	11
CHARACTERISTICS OF STUDIES	13
DATA AND ANALYSES	22
Analysis 1.1. Comparison 1 Terlipressin alone or with albumin versus no intervention or albumin, Outcome 1 Mortality.	23
Analysis 1.2. Comparison 1 Terlipressin alone or with albumin versus no intervention or albumin, Outcome 2 Reversal of	
hepatorenal syndrome	24
Analysis 1.3. Comparison 1 Terlipressin alone or with albumin versus no intervention or albumin, Outcome 3 Improved	
renal function	25
Analysis 1.4. Comparison 1 Terlipressin alone or with albumin versus no intervention or albumin, Outcome 4 Adverse	
events.	26
Analysis 1.5. Comparison 1 Terlipressin alone or with albumin versus no intervention or albumin, Outcome 5 Mortality in	
trials on type 1 hepatorenal syndrome	28
Analysis 1.6. Comparison 1 Terlipressin alone or with albumin versus no intervention or albumin, Outcome 6 Mortality in	
trials on terlipressin and albumin	29
Analysis 1.7. Comparison 1 Terlipressin alone or with albumin versus no intervention or albumin, Outcome 7 Mortality in	
trials with a low risk of selection bias	30
Analysis 1.8. Comparison 1 Terlipressin alone or with albumin versus no intervention or albumin, Outcome 8 Mortality	
analysed using a fixed effect model	31
APPENDICES	31
WHAT'S NEW	33
CONTRIBUTIONS OF AUTHORS	33
DECLARATIONS OF INTEREST	34
SOURCES OF SUPPORT	34
DIFFERENCES BETWEEN PROTOCOL AND REVIEW	34
INDEX TERMS	34

#### [Intervention Review]

## Terlipressin for hepatorenal syndrome

Lise Lotte Gluud<sup>1</sup>, Kurt Christensen<sup>2</sup>, Erik Christensen<sup>3</sup>, Aleksander Krag<sup>4</sup>

<sup>1</sup>Department of Internal Medicine, Gentofte University Hospital, Hellerup, Denmark. <sup>2</sup>Cochrane Hepato-Biliary Group, Copenhagen Trial Unit, Centre for Clinical Intervention Research, Department 3344, Rigshospitalet, Copenhagen University Hospital, Copenhagen, Denmark. <sup>3</sup>Clinic of Internal Medicine I, Bispebjerg Hospital, Copenhagen, Denmark. <sup>4</sup>Department of Gastroenterology 360, Copenhagen University Hospital Hvidovre, Hvidovre, Denmark

Contact address: Lise Lotte Gluud, Department of Internal Medicine, Gentofte University Hospital, Niels Andersensvej 65, Hellerup, 2900, Denmark. liselottegluud@yahoo.dk.

Editorial group: Cochrane Hepato-Biliary Group.

Publication status and date: New search for studies and content updated (no change to conclusions), published in Issue 9, 2012. Review content assessed as up-to-date: 31 January 2012.

Citation: Gluud LL, Christensen K, Christensen E, Krag A. Terlipressin for hepatorenal syndrome. *Cochrane Database of Systematic Reviews* 2012, Issue 9. Art. No.: CD005162. DOI: 10.1002/14651858.CD005162.pub3.

Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

## ABSTRACT

#### Background

Clinical trials suggest that terlipressin improves renal function in hepatorenal syndrome, but the evidence concerning mortality is equivocal.

#### Objectives

To assess the beneficial and harmful effects of terlipressin alone or with albumin versus placebo, no intervention or albumin for hepatorenal syndrome.

#### Search methods

Eligible trials were identified through electronic (*The Cochrane Library*, MEDLINE, EMBASE and Science Citation Index databases) and manual searches until January 2012.

#### Selection criteria

Randomised clinical trials involving patients with type 1 or type 2 hepatorenal syndrome were included irrespective of publication status or language.

#### Data collection and analysis

The review authors independently extracted data from trial reports and undertook correspondence with the authors. Primary outcome measures included mortality, reversal of hepatorenal syndrome and adverse events. Intention-to-treat, random-effects model metaanalyses were performed and results were expressed as risk ratios (RR) with 95% confidence intervals (CI), and the I<sup>2</sup> statistic provided a measure of intertrial heterogeneity. Subgroup, sensitivity, regression and sequential analyses were performed.

#### Main results

We identified six randomised clinical trials. All had high risk of bias. Five trials assessed terlipressin (with albumin in three trials) versus no intervention (with albumin in three trials) and one trial assessed terlipressin versus albumin. Data from five randomised trials on terlipressin alone (one trial) or terlipressin and albumin (four trials) were included in the review. In total, 74 of 155 (47.7%) patients randomised to terlipressin alone or terlipressin with albumin versus 98 of 154 (63.6%) patients randomised to no intervention, placebo

or albumin died. Random-effects model meta-analysis found that terlipressin reduced mortality (RR 0.76, 95% CI 0.61 to 0.95). The results were stable when repeated with trials on terlipressin plus albumin, trials on patients with type 2 hepatorenal syndrome, and trials with a low risk of selection bias. No evidence of bias or small study effects were identified in regression analyses. In a trial sequential analysis on mortality, the cumulative Z curve approached but did not cross the monitoring boundary suggesting that the results were not stable to adjustment for sparse data and multiple comparisons. Analyses of the remaining outcome measures found that terlipressin and albumin increased the number of patients with reversal of hepatorenal syndrome as well as adverse events, including cardiovascular and gastrointestinal symptoms.

#### Authors' conclusions

Terlipressin may reduce mortality and improve renal function in patients with type 1 hepatorenal syndrome. Whether the evidence is strong enough to support the intervention for clinical practice could be debated due to the results of the trial sequential analyses. However, the outcome measures assessed are objective, which reduces the risk of bias.

### PLAIN LANGUAGE SUMMARY

#### Terlipressin for patients with hepatorenal syndrome

Patients with severe cirrhosis of the liver may develop kidney failure. The disease is known as hepatorenal (liver-kidney) syndrome. The syndrome is divided into two types, type 1 has a rapid course of the disease whereas type 2 has a more protracted course. The disease may develop as a consequence of the circulatory changes that are associated with cirrhosis. Untreated, the disease is associated with high mortality. The median survival ranges between two weeks to six months. Terlipressin is a drug that affects the circulation and may help reverse the circulatory changes that lead to hepatorenal syndrome.

The present review includes data from five randomised trials on terlipressin alone or with placebo, no intervention or albumin. Our analyses suggest that terlipressin with albumin reduces mortality and improves renal function. The intervention increases the risk of adverse events.

## BACKGROUND

#### **Description of the condition**

Hepatorenal syndrome is a potentially reversible renal failure associated with severe liver disease (Arroyo 1996). The disease is relatively common among patients with decompensated cirrhosis. In a cohort study of 234 non-azotaemic patients with cirrhosis and ascites, 18% developed hepatorenal syndrome after one year (Gines 1993). The diagnosis includes cirrhosis and ascites plus impaired renal function after exclusion of parenchymal renal disease and factors that may precipitate renal dysfunction in cirrhosis (Salerno 2007). Hepatorenal syndrome is divided into two types with type 1 having the most rapid course of development. Without treatment, type 1 has a median survival of about two weeks and type 2 a median survival of about six months (Arroyo 1996; Gines 2003; Salerno 2007).

#### **Description of the intervention**

The development of hepatorenal syndrome has been associated with the circulatory changes seen in cirrhosis of the liver subsequent to portal hypertension and vasodilation of the splanchnic arteries (Cardenas 2003). The vasodilation results in effective underfilling of the renal arteries and activation of the renin-angiotensin-aldosterone, the arginine-vasopressin, and the sympathetic nervous systems (Pasqualetti 1998; Moller 2004; Ruiz del Arbol 2005). Activation of these systems may in turn lead to severe vasoconstriction of the renal arteries and hepatorenal syndrome (Cardenas 2003). The process may be reversed by vasoactive drugs that increase the splanchnic arterial tone. Vasopressin was evaluated initially but was abandoned as it led to severe ischaemias of the mesenteric mucosa, skin and myocardium (Obritsch 2004). Terlipressin is a vasopressin analogue which was introduced as a safer alternative (Freeman 1982). Subsequent studies have evaluated the effects of the somatostatin analogue octreotide and similar vasoconstrictors (Arroyo 2000). The initial trials on terlipressin

were small and some used a cross-over design with a short treatment duration and length of follow-up (Hadengue 1998; Solanki 2003). Subsequent trials have been larger but did not find convincing effects on clinical outcome measures (Martín-Llahí 2008; Neri 2008; Sanyal 2008).

#### Why it is important to do this review

Meta-analyses on terlipressin for hepatorenal syndrome have been conducted but with equivocal findings (Fabrizi 2009; Dobre 2010; Sagi 2010). Since large randomised trials were published after our previous review (Cochrane 2006), we performed the present systematic review with an update of the available evidence.

## OBJECTIVES

The primary objective was to assess the beneficial and harmful effects of terlipressin for hepatorenal syndrome.

## METHODS

### Criteria for considering studies for this review

#### **Types of studies**

Randomised trials were included irrespective of blinding, publication status or language. Trials using a cross-over design were included if data from the first period could be obtained.

#### **Types of participants**

Patients described as having type 1 or type 2 hepatorenal syndrome were included.

#### **Types of interventions**

The analyses included comparisons of terlipressin alone or with albumin versus placebo, no intervention or albumin.

#### Types of outcome measures

Based on the guidelines of the Cochrane Hepato-Biliary Group (CHBG), all outcome measures were were classed as primary outcomes. Furthermore, quality of life was added as an outcome measure.

Terlipressin for hepatorenal syndrome (Review)

Copyright © 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

#### **Primary outcomes**

Based on the Cochrane Hepato-Biliary Group guidelines and peer review comments, all outcome measures were classed as primary.

- Mortality (all-cause)
- Morbidity: reversal of hepatorenal syndrome and improved
- renal function as defined by the authors of included trials
  - Quality of life
  - Adverse events

#### Search methods for identification of studies

#### **Electronic searches**

Electronic searches were performed in *The Cochrane Library*, MEDLINE, EMBASE and Web of Science (Appendix 1).

#### Searching other resources

Manual searches included scanning of reference lists in relevant articles and conference proceedings. Additional ongoing or unpublished trials were identified through searches of trial registers through the World Health Organization (WHO) Clinical Trials Registry Platform search portal (http://apps.who.int/trialsearch/).

### Data collection and analysis

#### Selection of studies

Two authors (LG and AK) independently selected trials eligible for inclusion from the updated literature searches. Excluded trials were listed with the reasons for exclusion.

#### Data extraction and management

Review authors independently extracted data (LG and AK). All disagreements were resolved through discussion. We wrote to authors of the included trials to obtain additional information not described in the published reports.

#### Assessment of risk of bias in included studies

The allocation methods were extracted as the primary methods of bias control (Wood 2008). The assessment included the allocation sequence generation (classed as low risk of bias if based on a table of random numbers, computer generated random numbers or similar) and the allocation concealment (classed as low risk of bias if patients were randomised through a central independent unit, serially numbered opaque sealed envelopes or similar). We also assessed whether trials were described as double blind and the method of blinding (with double blind placebo controlled trials classed as low risk of bias), whether incomplete outcome data were addressed (classed as low risk of bias if all patients were accounted for in the report and analysis), whether the clinically relevant outcome measures were defined and reported (selective reporting), and other potential biases including sample size calculations and whether the trial was terminated early or was prolonged. For trials terminated early, we recorded whether this was based on predefined criteria.

#### Measures of treatment effect

All outcome measures were dichotomous and were expressed using relative risks (RR) with 95% confidence intervals (CI).

#### Unit of analysis issues

Since our primary outcome measure was mortality, and since hepatorenal syndrome is a disease with a fluctuating course, we only planned to include data from the first treatment period of crossover trials, if any had been identified.

#### Dealing with missing data

Data were sought on all patients randomised to allow intentionto-treat analyses. For patients with missing outcome data, carry forward of the last observed response was used.

#### Assessment of heterogeneity

Intertrial heterogeneity was expressed using the I<sup>2</sup> statistic.

#### Assessment of reporting biases

Evidence of reporting bias was assessed as described above. We also planned to compare outcome measures described in protocols and published reports, if any protocols had been available.

#### Data synthesis

The analyses were performed using the statistical programs RevMan 5 (RevMan), STATA (STATA) and TSA (TSA 2008). All meta-analyses were performed using random-effects models due to expected clinical heterogeneity (based on patient inclusion criteria and treatment dose plus duration).

#### Subgroup analysis and investigation of heterogeneity

For all-cause mortality, we performed subgroup regression and sequential analyses to evaluate sources of intertrial heterogeneity, bias, small study effects and errors associated with cumulative testing (Higgins 2011). In the subgroup analyses, we analysed the effect of the intervention when including only:

- trials on type 1 hepatorenal syndrome;
- trials on terlipressin with albumin;
- trials with a low risk of selection bias based on the
- assessment of allocation methods.

The risk of small study effects was assessed through regression analyses (Egger's test).

Sequential analyses were performed using the monitoring boundaries approach (Wetterslev 2008; Higgins 2011). The analysis was performed with alpha set to 5%, power to 80%, model-based heterogeneity, and including the control event proportion and intervention effects observed in the meta-analysis. The effect measure was relative risk (random-effects model), the relative risk reduction 25% (low bias based), incidence in the control arm of 76%, and heterogeneity correction 47%.

#### Sensitivity analysis

The sensitivity and subgroup analyses included repetitive metaanalyses using a fixed-effect model to afford less weight to smaller trials.

## RESULTS

#### **Description of studies**

See: Characteristics of included studies; Characteristics of excluded studies; Characteristics of ongoing studies.

#### **Results of the search**

After scanning the 345 potentially eligible references identified in the literature searches (Appendix 1), 20 references referred to trials that were potentially eligible for inclusion (Figure 1). Among these, nine that referred to randomised trials on patients with hepatorenal syndrome had to be excluded because they did not evaluate the interventions assessed in the present review. The remaining 11 references referred to six randomised trials that fulfilled our inclusion criteria (Hadengue 1998; Yang 2001; Solanki 2003; Martín-Llahí 2008; Neri 2008; Sanyal 2008). One trial used a cross-over design (Hadengue 1998). We were unable to retrieve data from the first period of the trial. Accordingly, the trial could not be included in our quantitative analyses. Since no data were included from the cross-over trial, the trial is not included in the description of included studies.



Figure I. Study flow diagram.

Terlipressin for hepatorenal syndrome (Review) Copyright 0 2012 The Cochrane Collaboration. Published by John Wiley & Sons, Ltd.

#### **Included studies**

#### **Trial characteristics**

All trials were published as full paper articles (Characteristics of included studies). One trial was translated from Chinese (Yang 2001). The remaining trials were published in English. The trials were conducted in the United States, Italy, Spain, Canada, India, China, Germany and Russia. All trials were performed in specialised units in an intensive or semi-intensive setting. Two trials were multicentred and three were single centre trials. The included patients were followed for six months in one trial (Sanyal 2008) and to the end of treatment in four trials. None of the included trials assessed health economics.

#### **Patient characteristics**

Hepatorenal syndrome was diagnosed based on evidence of cirrhosis, serum creatinine more than 133 µmol/L (1.5 mg/dL) after diuretic withdrawal and volume expansion plus absence of shock, ongoing infections, parenchymal renal disease, and treatment with nephrotoxic drugs. In one trial (Martín-Llahí 2008) patients were diagnosed as having type 2 hepatorenal syndrome based on a serum creatinine more than 175 µmol/L (1.97 mg/dl) and absence of bacterial infection associated with findings of a systemic inflammatory response. In the remaining trials the type of hepatorenal syndrome was diagnosed based on the course of the disease, with type 1 developing within two weeks (Arroyo 1996). One trial did not report the proportion of patients with type 1 hepatorenal syndrome (Yang 2001). In one trial (Martín-Llahí 2008), 56% of patients had type 1 and 44% had type 2 hepatorenal syndrome. In remaining trials, all patients had type 1 hepatorenal syndrome at baseline. The mean age in the treatment and control groups ranged from 51 to 59 years and 52 to 60 years, respectively, The

proportion of men ranged from 40% to 71% and the proportion with alcoholic liver disease from 13% to 72%.

#### Intervention characteristics

The median initial dose of terlipressin was 1 mg four times daily. In two trials (Yang 2001; Solanki 2003) the dose of terlipressin was not adjusted. In the remaining trials, the dose of terlipressin was increased to 2 mg four to six times daily after three days in the non-responders (patients without improved renal function). One trial assessed terlipressin versus no intervention (Yang 2001); one trial assessed terlipressin versus placebo (Hadengue 1998); one trial assessed terlipressin versus albumin (Neri 2008); and three trials assessed terlipressin plus albumin verus albumin (Solanki 2003; Martín-Llahí 2008; Sanyal 2008). Accordingly, five trials assessed terlipressin versus no intervention or placebo (Hadengue 1998; Yang 2001; Solanki 2003; Martín-Llahí 2008; Sanyal 2008). Of the former five trials, three used albumin both in the experimental group and in the control group.

#### **Excluded studies**

None of the excluded trials assessed the interventions specified for the present review. The excluded trials compared terlipressin and albumin versus other vasoactive drugs for hepatorenal syndrome (Excluded studies).

#### **Risk of bias in included studies**

#### Allocation

One trial did not report the allocation methods. This trial was classed as having unclear control of selection bias (Figure 2). The remaining trials reported adequate allocation and were classed as having a low risk of selection bias.



Figure 2. Methodological quality summary: review authors' judgments about each methodological quality item for each included study.

#### Blinding

One trial reported double blinding of patients and investigators by use of a placebo infusion (Figure 2). One trial was described as single blind. The trial did not specify whether blinding was maintained for patients, investigators, or other persons involved in the trial. Four trials were conducted without blinding.

#### Incomplete outcome data

Three trials reported dropouts and withdrawals and included all patients in intention-to-treat analyses (Martín-Llahí 2008; Neri 2008; Sanyal 2008). One trial followed patients to the end of treatment but obtained additional follow-up data for some of the included patients (Solanki 2003). The strategy for obtaining additional information and the number of patients without additional follow-up data was not reported. In the remaining trials, no losses to follow-up were reported.

#### Selective reporting

All trials reported clinically relevant outcome measures.

Other potential sources of bias

One trial reported sample size calculations and achieved the required sample size (Sanyal 2008). One trial was terminated prematurely due to unexpectedly low event rates (Martín-Llahí 2008). The remaining trials did not report sample size calculations or whether the trials were terminated early.

#### **Effects of interventions**

### Mortality

Random-effects model meta-analysis found that terlipressin, alone or with albumin, reduced mortality compared with no intervention, placebo or albumin (RR 0.75, 95% CI 0.59 to 0.97;  $I^2$ 39%; Analysis 1.1). The effect was also seen when the analysis was limited to trials on patients with type 1 hepatorenal syndrome (Analysis 1.5), trials on terlipressin and albumin (Analysis 1.6), and trials with a low risk of selection bias (Analysis 1.7). Regression analysis showed no evidence of bias (P = 0.224). In trial sequential analysis, the cumulative Z curve approached but did not cross the monitoring boundary suggesting that the results were not stable to adjustments for sparse data and multiple testing (Figure 3). A fixed-effect model meta-analysis found that terlipressin plus albumin reduced mortality compared with no intervention or albumin (RR 0.76, 95% CI 0.63 to 0.91; Analysis 1.8).

Figure 3. Trial sequential analysis of terlipressin versus no intervention on mortality in patients with hepatorenal syndrome. The required information size was calculated to 466 participants based upon a control group mortality of 76%; a relative risk reduction of 25%; an alpha of 5%; and a beta of 20%. The blue cumulative Z curve crosses the conventional alpha of P = 0.05 twice, but not the trial sequential alpha-spending monitoring boundaries (inward sloping red lines).



#### **Reversal of hepatorenal syndrome**

All trials reporting the number of patients with reversal of hepatorenal syndrome and improved renal function assessed terlipressin and albumin versus albumin (Analysis 1.2). Random-effects model meta-analysis showed that terlipressin had a beneficial effect on this outcome measure.

#### Quality of life

None of the included trials assessed the quality of life.

#### Adverse events

Terlipressin also increased the risk of cardiovascular adverse events (RR 7.26, 95% CI 1.70 to 31.05;  $I^2 = 0\%$ ; Analysis 1.4). The remaining adverse events included abdominal pain, diarrhoea, bacterial infections, chest pain, circulatory overload, gastrointestinal bleeding, hepatic encephalopathy, livedo reticularis, respiratory distress or acidosis and did not appear to differ between the intervention and control groups.

## DISCUSSION

#### Summary of main results

The present review suggests that terlipressin may reduce mortality and have a beneficial effect on renal function in type 1 hepatorenal syndrome. However, most of the patients were only followed to the end of treatment and there are few available data on adverse events. Accordingly, the prognosis of the underlying liver disease should be considered in treatment decisions.

The evidence on the use of terlipressin alone and intervention benefits in type 2 hepatorenal syndrome were scarce. Only one of the included trials assessed terlipressin alone (Yang 2001). The trial was small and the findings were inconclusive. Likewise, only one of the trials on terlipressin and albumin versus albumin included patients with type 2 hepatorenal syndrome (Martín-Llahí 2008). The number of patients with type 2 hepatorenal syndrome included in the trial was relatively small. Accordingly, no clear intervention effects were identified for this patient group. There was, however, clear evidence suggesting that treatment with terlipressin is associated with a number of adverse events. In particular, the intervention seems to increase the risk of cardiovascular adverse events, some of which may be potentially serious. Other patient reports also show that terlipressin may be associated with severe

adverse effects (Shawcross 2004; Krag 2008). The intervention should, therefore, be closely monitored.

The present review includes three small trials comparing terlipressin plus albumin versus albumin alone, in different dosing regimens (continuous versus bolus administration). The fact that no clear intervention effects were identified may reflect that the reported results were interim analyses of part of the planned patient cohort, although this was not specifically stated. We also identified three trials assessing terlipressin versus noradrenaline (Alessandria 2007a; Angeli 2008a; Sharma 2008a). The trials on terlipressin versus noradrenaline also found no clear differences between the intervention groups. The trials did not report sample size calculations but, based on the number of patients included, the trials were not designed to establish equivalence. Additional evidence is needed to reassess the results.

## Overall completeness and applicability of evidence

We found little evidence of clinical intertrial heterogeneity. The mean control group Child-Pugh scores were remarkably similar (11 in three trials (Martín-Llahí 2008; Neri 2008; Sanyal 2008)). Likewise, the included trials on terlipressin and albumin used similar criteria to diagnose hepatorenal syndrome, based on previous recommendations (Arroyo 1996). The current diagnostic criteria include presence of cirrhosis, ascites, elevated serum creatinine after at least 48 hours of diuretic withdrawal and volume expansion combined with absence of shock, treatment with nephrotoxic drugs, and parenchymal renal disease (Salerno 2007). The use of minor criteria and exclusion of patients with infections is now omitted from the criteria. Type 1 hepatorenal syndrome is now defined as renal failure with serum creatinine increasing to 226 µmol/L (2.5 mg/dL) within two weeks. Type 2 hepatorenal syndrome is defined as a moderate to slowly progressive renal failure with serum creatinine between 133 and 226 µmol/L (1.5 to 2.5 mg/dL). Although the included trials used previously established criteria, the evidence is likely to be applicable today. It may, however, be argued that there is still room for trials on terlipressin and albumin using the current diagnostic criteria.

The duration of the effect of terlipressin on mortality should be considered when deciding whether or not to treat a patient with hepatorenal syndrome (Gluud 2010). Some patients may die in spite of a clear improvement in renal function (Martín-Llahí 2008; Sanyal 2008). After an initial complete normalisation of renal function, hepatorenal syndrome may reappear. We attempted to perform a post hoc analysis to determine the effect of treatment on recurrence of hepatorenal syndrome but we were unable to extract the necessary data.

#### Quality of the evidence

The present review identified a number of methodological concerns, including unclear randomisation, lack of sample size calculations and lack of blinding. Previous meta-epidemiological studies show that allocation concealment is one of the most important predictors of bias control (Wood 2008). The allocation sequence generation and allocation concealment were classed as adequate in all trials on terlipressin and albumin versus albumin. Excluding the trial without adequate randomisation had no influence on the overall results.

Another potential source of bias in included trials was related to the lack of reported sample size calculations. For the trials without sample size calculations, we were unable to determine whether the trials were terminated prematurely, terminated at an arbitrary point, or extended due to low event rates. One of the included trials reporting sample size calculations had to be terminated prematurely due to unexpectedly low event rates (Martín-Llahí 2008). The trial assessed terlipressin plus albumin versus albumin and was terminated after an interim analysis suggested that 2000 patients would be required to achieve sufficient statistical power. Whether the interim results reflect a true (low) intervention effect, a random error, or the inclusion criteria is difficult to assess. One possible explanation could be that a number of the included patients had type 2 hepatorenal syndrome. Overall, there is little evidence on this latter patient group.

#### Potential biases in the review process

One of the main limitations of the present review is related to the relatively low overall sample size. Identification of patients who clearly fulfil the diagnostic criteria for hepatorenal syndrome may be difficult, as is the recruitment of critically ill patients in clinical trials. Accordingly, the largest trials were multicentred and multinational (Sanyal 2008). This involvement of several clinical sites in more than one geographical region increases the clinical heterogeneity. On the other hand, the heterogeneity also increases the external validity, making it possible to extrapolate the results to larger patient populations in similar specialised centres. The heterogeneity increases the need for additional subgroup and sensitivity analyses. Analysis of individual patient data would have increased the possibilities of performing such analyses. Unfortunately, we were not able to conduct individual patient data metaanalyses, and we were only able to perform a few additional analyses to explore sources of intertrial heterogeneity due to the limited number of included trials.

## Agreements and disagreements with other studies or reviews

Three of the included trials found that baseline serum creatinine was an independent predictor of survival (Martín-Llahí 2008; Neri 2008; Sanyal 2008). In our analyses, the baseline creatinine in the

Terlipressin for hepatorenal syndrome (Review)

control groups of the trials on terlipressin plus albumin ranged from 194 to 362  $\mu$ mol/L (2.2 to 4.1 mg/dL). All trials found similar baseline values for the treatment and control groups. In agreement with previous findings, our analyses suggest that the treatment effect was the largest in the trial with the lowest baseline serum creatinine (Solanki 2003). This may suggest that treatment should be administered early and that a protracted deterioration in renal function impedes recovery.

A number of meta-analyses have assessed the effect of terlipressin for hepatorenal syndrome (Fabrizi 2009; Dobre 2010; Sagi 2010). The results concerning mortality are equivocal. In agreement with our findings, one meta-analysis found that terlipressin increases survival among patients with type 1 hepatorenal syndrome (Sagi 2010). The two remaining meta-analyses found no clear effect of terlipressin on survival, although only one performed a meta-analvsis addressing this question (Fabrizi 2009). In agreement with our findings, all reviews found that terlipressin improves renal function but also increases the risk of cardiovascular and ischaemic adverse events. The differences between the conclusions in the different reviews are mainly related to the inclusion criteria. For example, one review only included placebo controlled trials (Fabrizi 2009). This decision is not clearly supported by previous evidence on the importance of bias control in randomised trials (Gluud 2006; Wood 2008). Although lack of blinding may affect the risk of bias, there is no clear or consistent evidence to support the exclusion of open trials from meta-analyses since the effect of blinding is inconsistent across trials. The extent as well as the effect of bias associated with lack of blinding is unpredictable and does not support the a priori exclusion of trials based on this component alone.

## AUTHORS' CONCLUSIONS

#### Implications for practice

The combined evidence suggests that terlipressin may be considered for patients with type 1 hepatorenal syndrome.

#### Implications for research

Additional research may be needed to assess the effect of terlipressin for hepatorenal syndrome that is m identified by the updated diagnostic criteria. Future trials may need to be planned at a multinational level since inclusion of a sufficient number of patients may otherwise prove difficult.

## ACKNOWLEDGEMENTS

Thank you to the authors who provided additional information on the included trials. We also thank Sarah Klingenberg from the Cochrane Hepato-Biliary Group for designing the electronic searches and Maette Kjaer who participated in a previous version of this review but was unable to continue due to other commitments.

Contact Editors: Gennaro D'Amico, Italy; Ronald L Koretz, USA; Christian Gluud, Denmark.

#### REFERENCES

#### References to studies included in this review

#### Hadengue 1998 {published data only}

Hadengue A, Gadano A, Giostra E, Negro F, Moreau R, Valla D, et al. Terlipressin in the treatment of hepatorenal syndrome (HRS): a double blind cross-over study. *Hepatology* 1995;**22**:165A.

\* Hadengue A, Gadano A, Moreau R, Giostra E, Durand F, Valla D, et al.Beneficial effects of the 2-day administration of terlipressin in patients with cirrhosis and hepatorenal syndrome. *Journal of Hepatology* 1998;**29**:565–70. [MEDLINE: PMID: 9824265]

#### Martín-Llahí 2008 {published data only}

Martin-Llahi M, Pepin MN, Guevara M, Torre A, Monescillo A, Soriano G, et al.Randomized, comparative study of terlipressin and albumin vs albumin alone in patients with cirrhosis and hepatorenal syndrome. *Journal* of *Hepatology* 2007;**46**:S36.

\* Martín-Llahí M, Pépin MN, Guevara M, Díaz F, Torre A, Monescillo A, et al. Terlipressin and albumin vs albumin in patients with cirrhosis and hepatorenal syndrome: a randomized study. *Gastroenterology* 2008;**134**:1352–9. [: ClnicalTrials.gov NCT00287664]

#### Neri 2008 {published data only}

Neri S, Pulvirenti D, Malaguarnera M, Cosimo BM, Bertino G, Ignaccolo L, et al. Terlipressin and albumin in patients with cirrhosis and type I hepatorenal syndrome. *Digestive Diseases and Sciences* 2008;**53**:830–5. [PUBMED: 17939047]

#### Sanyal 2008 {published data only}

Sanyal A, Boyer T, Garcia-Tsao G, Regenstein F, Rossaro L, Teuber P. A prospective, randomized, double blind, placebo-controlled trial of terlipressin for type 1 hepatorenal syndrome (HRS). *Hepatology* 2006;**44**:694A.

\* Sanyal AJ, Boyer T, Garcia-Tsao G, Regenstein F, Rossaro L, Appenrodt B, et al.A randomized, prospective, doubleblind, placebo-controlled trial of terlipressin for type 1 hepatorenal syndrome. *Gastroenterology* 2008;**134**: 1360–8. [MEDLINE: 18471513; : clinicaltrials.gov NCT00089570] Sanyal AJ, Boyer TD, Garcia-Tsao G, Blei AT, Teuber P,

Terlipressin for hepatorenal syndrome (Review)

Terlipressin Study Group. Effect of terlipressin on mean arterial pressure (MAP) and its relation to serum creatinine concentration in type ! hepatorenal syndrome (HRS): analysis of data from the pivotal phase 3 trial. *Hepatology* 2008;**48**:1071A.

Sanyal AJ, Boyer TD, Teuber P. Prognostic factors for hepatorenal syndrome (HRS) in patients with type 1 HRS enrolled in a randomized double-blind, placebo-controlled trial. *Hepatology* 2007;**46**:565A.

#### Solanki 2003 {published data only}

Solanki P, Chawla A, Garg R, Gupta R, Jain M, Sarin SK. Beneficial effects of terlipressin in hepatorenal syndrome: a prospective, randomized placebo-controlled clinical trial. *Journal of Gastroenterology and Hepatology* 2003;**18**:152–6. [MEDLINE: PMID 12542598]

#### Yang 2001 {published data only}

Yang YZ, Dan ZL, Liu NZ, Liu M, et al. [Efficacy of terlipressin in treatment of liver cirrhosis with hepatorenal syndrome]. *Journal of Internal Intensive Medicine* 2001;7: 123–5.

#### References to studies excluded from this review

#### Alessandria 2007 {published data only}

Alessandria C, Marzano A, Ottobrelli A, Debernardi-Venon W, Todros L, Torrni M, et al.Noradrenaline vs terlipressin in hepatorenal syndrome: a prospective, randomized study. *Journal of Hepatology* 2006;44:S83.

\* Alessandria C, Ottobrelli A, Debernardi-Venon W, Todros L, Cerenzia MT, Martini S, et al.Noradrenalin vs terlipressin in patients with hepatorenal syndrome: a prospective, randomized, unblinded, pilot study. *Journal of Hepatology* 2007;**47**:499–505. [: clinicaltrials.gov NCT00370253]

#### Angeli 2008 {published data only}

Angeli P, Fasolato S, Cavallin M, Maresio G, Callegaro A, Sticca A, et al. Terlipressin given as continuous intravenous infusion is the more suitable schedule for the treatment of type 1 hepatorenal syndrome (HRS) in patients with cirrhosis: results of a controlled clinical study. *Hepatology* 2008;**48**:378A.

#### Cavalli 2012 {published data only}

Cavallin M, Merli M, FasolatoS, Toniutto P, Salerno F, Bernardi M, et al. Terlipressin and albumin vs midodrine plus octreotide in the treatment of hepatorenal syndrome in patients with cirrhosis: results of a controlled trial by the Italian association for the study of the liver. *Digestive and Liver Disease* 2012;44:S10.

#### Chelarescu 2003 {published data only}

Chelarescu D, Chelarescu O, Crumpie F, Staratan I. Captopril in low dose associated with octreotide in hepatorenal syndrome: A randomized study. *Journal of Hepatology* 2003;**38**:S56.

#### Pomier 2003 {published data only}

Pomier-Layrargues G, Paquin SC, Hassoun Z, Lafortune M, Tran A. Octreotide in hepatorenal syndrome: a randomized, double-blind, placebo-controlled, crossover study. *Hepatology* 2003;**38**:238–43.

#### Sharma 2008 {published data only}

\* Sharma P, Kumar A, Shrama BC, Sarin SK. An open label, pilot, randomized controlled trial of noradrenaline versus terlipressin in the treatment of type 1 hepatorenal syndrome and predictors of response. *American Journal of Gastroenterology* 2008;**103**:1689–97.

Sharma P, Kumar A, Shrama BC, Sarin SK. Noradrenaline versus terlipressin in the treatment of type 1 hepatorenal syndrome: a randomzied controlled trial. *Hepatology* 2006; **44**:449A.

#### Silawat 2011 {published data only (unpublished sought but not used)}

Silawat FN, Shaikh MK, Lohana RK, Devrajani BR, Shah SZA, Ansari A. Efficacy of terlipressin and albumin in the treatment of hepatorenal syndrome. *World Applied Sciences Journal* 2011;**12**(11):1946–59.

#### References to ongoing studies

#### NCT00742339 {unpublished data only}

Terlipressin + albumin versus midodrine + octreotide in the treatment of hepatorenal syndrome.. Ongoing study May 2005..

#### NCT01143246 {unpublished data only}

A Placebo-Controlled, Double-Blind Study to Confirm the Reversal of Hepatorenal Syndrome Type 1 With Terlipressin.. Ongoing study September 2010..

#### Additional references

#### Arroyo 1996

Arroyo V, Gines P, Gerbes AL, Dudley FJ, Gentilini P, Laffi G, et al.Definition and diagnostic criteria of refractory ascites and hepatorenal syndrome in cirrhosis. International Ascites Club. *Hepatology* 1996;**23**:164–76.

#### Arroyo 2000

Arroyo V, Jimenez W. Complications of cirrhosis. II. Renal and circulatory dysfunction. Lights and shadows in an important clinical problem. *Journal of Hepatology* 2000;**32**: 157–70.

#### Cardenas 2003

Cardenas A, Arroyo V. Hepatorenal syndrome. *Annals of Hepatology* 2003;2:23–9.

#### Dobre 2010

Dobre M, Demirjian S, Sehgal AR, Navaneethan SD. Terlipressin in hepatorenal syndrome: a systematic review and meta-analysis. *International Urology and Nephrology* 2011;**43**(1):175–84.

#### Fabrizi 2009

Fabrizi F, Dixit V, Messa P, Martin P. Terlipressin for hepatorenal syndrome: A meta-analysis of randomized trials. *International Journal of Artificial Organs* 2009;**32**: 133–40.

#### Freeman 1982

Freeman JG, Cobden I, Lishman AH, Record CO. Controlled trial of terlipressin ('Glypressin') versus vasopressin in the early treatment of oesophageal varices. *Lancet* 1982;**2**:66–8.

#### Terlipressin for hepatorenal syndrome (Review)

#### Gines 1993

Gines A, Escorsell A, Gines P, Salo J, Jimenez W, Inglada L, et al.Incidence, predictive factors, and prognosis of hepatorenal syndrome in cirrhosis. *Gastroenterology* 1993; **105**:229–36. [MEDLINE: PMID: 8514039]

#### Gines 2003

Gines P, Guevara M, Arroyo V, Rodes J. Hepatorenal syndrome. *Lancet* 2003;**362**:1819–27.

#### Gluud 2006

Gluud LL. Bias in clinical intervention research. *American Journal of Epidemiology* 2006;**163**:493–501.

#### Higgins 2011

Higgins JP, Whitehead A, Simmonds M. Sequential methods for random-effects meta-analysis. *Statistics in Medicine* 2011;**30**(9):903–21. [PUBMED: 21472757]

## Krag 2008

Krag A, Borup T, Møller S, Bendtsen F. Efficacy and safety of terlipressin in cirrhotic patients with variceal bleeding or hepatorenal syndrome. *Advances in Therapy* 2008;**25**: 1105–40.

#### Moller 2004

Moller S, Henriksen JH. Review article: pathogenesis and pathophysiology of hepatorenal syndrome--is there scope for prevention?. *Alimentary Pharmacology & Therapeutics* 2004;**20**:31–41.

#### Obritsch 2004

Obritsch MD, Bestul DJ, Jung R, Fish DN, MacLaren R. The role of vasopressin in vasodilatory septic shock. *Pharmacotherapy* 2004;**24**:1050–3.

#### Pasqualetti 1998

Pasqualetti P, Festuccia V, Collacciani A, Acitelli P, Casale R. Circadian rhythm of arginine vasopressin in hepatorenal syndrome. *Nephron* 1998;**78**:33–7.

#### RevMan

The Nordic Cochrane Center. Review Manager (RevMan), version 5.0. The Nordic Cochrane Center, 2008.

#### Ruiz del Arbol 2005

Ruiz-del-Arbol L, Monescillo A, Arocena C, Valer P, Gines P, Moreira V, et al.Circulatory function and hepatorenal syndrome in cirrhosis. *Hepatology* 2005;**42**:439–47.

## Sagi 2010

Sagi SV, Mittal S, Kasturi KS, Sood GK. Terlipressin therapy for reversal of type 1 hepatorenal syndrome: a meta-analysis of randomized controlled trials. *Journal of Gastroenterology* and Hepatology 2010;**25**:880–5.

### Salerno 2007

Salerno F, Gerbes A, Ginès P, Wong F, Arroyo V. Diagnosis, prevention and treatment of hepatorenal syndrome in cirrhosis. *Gut* 2007;**56**:1310–8.

#### Shawcross 2004

Shawcross DL, Davies NA, Mookerjee RP, Hayes PC, Williams R, Lee A, et al.Worsening of cerebral hyperemia by the administration of terlipressin in acute liver failure with severe encephalopathy. *Hepatology* 2004;**39**:471–5.

#### STATA

Stata Corp. STATA. Stata Corp, 2007.

#### TSA 2008

The Copenhagen Trial Unit. Centre for Clinical Intervention Research, Rigshospitalet, Copenhagen University Hospital, Copenhagen, Denmark. www.ctu.dk/ tsa. Trial Sequential Analysis. The Copenhagen Trial Unit. Centre for Clinical Intervention Research, Rigshospitalet, Copenhagen University Hospital, Copenhagen, Denmark. www.ctu.dk/tsa, 2008.

#### Wetterslev 2008

Wetterslev J, Thorlund K, Brok J, Gluud C. Trial sequential analysis may establish when firm evidence is reached in cumulative meta-analysis. *Journal of Clinical Epidemiology* 2008;**61**:64–75.

#### Wood 2008

Wood L, Egger M, Gluud LL, Schulz KF, Jüni P, Altman DG, et al.Empirical evidence of bias in treatment effect estimates in controlled trials with different interventions and outcomes: meta-epidemiological study. *BMJ* 2008; **336**:601–5.

#### References to other published versions of this review

#### Cochrane 2006

Gluud LL, Kjaer MS, Christensen E. Terlipressin for hepatorenal syndrome. *Cochrane Database of Systematic Reviews* 2006, Issue 4. [DOI: 10.1002/ 14651858.CD005162]

#### Gluud 2010

Gluud LL, Christensen K, Christensen E, Krag A. Systematic review of randomized trials on vasoconstrictor drugs for hepatorenal syndrome. *Hepatology* 2010;**51**: 576–84.

\* Indicates the major publication for the study

Terlipressin for hepatorenal syndrome (Review)

## CHARACTERISTICS OF STUDIES

## Characteristics of included studies [ordered by study ID]

## Hadengue 1998

Methods	- Single centre trial.
Participants	<ul> <li>Mean age</li> <li>terlipressin group 53 years.</li> <li>placebo group 53 years.</li> <li>Proportion of men 56%.</li> <li>Proportion with alcoholic liver disease 78%.</li> </ul>
Interventions	Terlipressin • 1 mg twice daily for two days.
Outcomes	- Duration of follow-up: end of treatment.
Country of origin	France.
Interventions	Terlipressin versus placebo.
Proportion with type 1 hepatorenal syn- drome	100%
Notes	Cross-over trial. The trial does not report outcome measures for the first period. The trial is therefore not included in our quantitative analyses

## Risk of bias

Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Unclear risk	Not described.
Allocation concealment (selection bias)	Low risk	Identical coded drug containers.
Blinding (performance bias and detection bias) All outcomes	Low risk	Double blinding achieved using terlipressin placebo.
Incomplete outcome data (attrition bias) All outcomes	High risk	Clinical outcome measures are not reported according to the intervention group or treatment period. Three patients were withdrawn and excluded from the analyses
Selective reporting (reporting bias)	High risk	Clinical outcome measures not accounted for.

Terlipressin for hepatorenal syndrome (Review)

## Hadengue 1998 (Continued)

Other bias	High risk	Sample size calculation not reported.

## Martín-Llahí 2008

Methods	- Multicentre trial.		
Participants	Mean age • terlipressin plus albumin group 59 years. • albumin group 52 years. - Proportion of men 63%. - Proportion with alcoholic liver disease 72%.		
Interventions	<ul> <li>Terlipressin <ul> <li>1 mg six times daily. If no improvement in renal function was observed, the dose was increased to 2 mg six times daily.</li> <li>Albumin <ul> <li>1 g/kg for 24 hours then 40 g/day adjusted according to the central venous pressure.</li> </ul> </li> </ul></li></ul>		
Outcomes	- Duration of follow-up: 3 months after treatment.		
Country of origin	Spain.	Spain.	
Interventions	Terlipressin plus albumin versus albumin for a maximum of 15 days		
Proportion with type 1 hepatorenal syn- drome	56%		
Notes	Full paper article.		
Risk of bias			
Bias	Authors' judgement	Support for judgement	
Random sequence generation (selection bias)	Low risk	Computer generated random numbers.	
Allocation concealment (selection bias)	Low risk	Central randomisation.	
Blinding (performance bias and detection bias) All outcomes	High risk	Open trial.	
Incomplete outcome data (attrition bias) All outcomes	Low risk	Intention-to-treat analyses with all patients randomised reported	

## Martín-Llahí 2008 (Continued)

Selective reporting (reporting bias)	Low risk	<ul> <li>Clinically relevant outcome measures reported.</li> <li>Primary outcome measures <ul> <li>mortality after three months and</li> <li>improvement in renal function.</li> </ul> </li> </ul>
Other bias	High risk	- Sample size calculations performed. The trial was terminated after a preliminary analysis suggested that the event rate was considerably lower than expected

## Neri 2008

Methods	- Multicentre trial.	
Participants	Mean age • terlipressin group 59 years • albumin group 60 years. - Proportion of men 40%. - Proportion with alcoholic liver disease 13%.	
Interventions	<ul> <li>Terlipressin</li> <li>1 mg four times daily for 5 days then 0.5 mg four times/day for 14 days.</li> <li>Albumin</li> <li>1 g/kg for 24 hours then 40 to 80 g/day.</li> </ul>	
Outcomes	- Duration of follow-up: six months after hospital discharge	
Country of origin	Italy.	
Interventions	Terlipressin plus albumin versus albumin for 19 days.	
Proportion with type 1 hepatorenal syn- drome	100%	
Notes	- Patients with recurrence of hepatorenal syndrome after the initial treatment were treated with terlipressin plus albumin	
Risk of bias		
Bias	Authors' judgement	Support for judgement
Random sequence generation (selection bias)	Low risk	Computer generated list of random numbers.
Allocation concealment (selection bias)	Low risk	Central randomisation.
Blinding (performance bias and detection bias) All outcomes	High risk	Open trial.

Incomplete outcome data (attrition bias) All outcomes	Low risk	No losses to follow-up reported.
Selective reporting (reporting bias)	Low risk	Clinically relevant outcome measures reported. Primary outcome measure • Resolution of hepatorenal syndrome defined as normalisation of creatinine.
Other bias	Unclear risk	Sample size calculation not reported.

Sanyal 2008

Methods	- Multicentre trial.		
Participants	<ul> <li>Mean age</li> <li>terlipressin plus albumin group 51 years.</li> <li>albumin group 53 years.</li> <li>Proportion of men 71%.</li> <li>Proportion with alcoholic liver disease 36%.</li> </ul>		
Interventions	<ul> <li>Terlipressin <ul> <li>1 mg four times daily increased to 2 mg four times daily if serum creatinine was not decreased by at least 30%</li> <li>Albumin <ul> <li>100 g for 24 hours then 25 g daily.</li> </ul> </li> </ul></li></ul>		
Outcomes	Follow-up: 6 months.		
Country of origin	United States, Germany, and Russia.		
Interventions	Terlipressin plus albumin versus albumin for a maximum of 14 days		
Proportion with type 1 hepatorenal syn- drome	100%		
Notes	- Full paper article.		
Risk of bias			
Bias	Authors' judgement	Support for judgement	
Random sequence generation (selection bias)	Low risk	Computer generated random numbers.	
Allocation concealment (selection bias)	Low risk	Central randomisation.	

## Sanyal 2008 (Continued)

Blinding (performance bias and detection bias) All outcomes	Low risk	Double blind using terlipressin placebo.
Incomplete outcome data (attrition bias) All outcomes	Low risk	Intention-to-treat analyses including all patients randomised are reported
Selective reporting (reporting bias)	Low risk	Clinically relevant outcome measures reported. Primary outcome measure • treatment success at day 14 defined as normalisation of serum creatinine on 2 measurements with at least 48 hour intervals and no dialysis, death, or recurrence of HRS type 1 before day 15.
Other bias	Low risk	Sample size calculation reported and preset sample size achieved

## Solanki 2003

Methods	- Single centre trial.		
Participants	Mean age • terlipressin and albumin group 51 years. • albumin group 52 years. - Proportion of men 71%. - Proportion with alcoholic liver disease 33%.		
Interventions	Terlipressin • 1 mg twice daily. Albumin • 20 g daily.		
Outcomes	- Duration of follow-up: end of treatment.		
Country of origin	India.		
Interventions	Terlipressin plus album	in versus placebo plus albumin for 15 days	
Proportion with type 1 hepatorenal syn- drome	100%		
Notes	Full paper article.		
Risk of bias			
Bias	Authors' judgement	Support for judgement	

## Solanki 2003 (Continued)

Random sequence generation (selection bias)	Low risk	Random number table	
Allocation concealment (selection bias)	Low risk	Centralised randomisation through an independent statistician	
Blinding (performance bias and detection bias) All outcomes	High risk	The trial is described as single blind, but whether the patient or investigators were blinded is not described	
Incomplete outcome data (attrition bias) All outcomes	Low risk	There were no losses to follow-up.	
Selective reporting (reporting bias)	Low risk	Clinically relevant outcome measures reported.	
Other bias	Unclear risk	Sample size calculation: not reported.	
Yang 2001			
Methods	Single centre trial.	Single centre trial.	
Participants	Patient characteristics not reported.		
Interventions	Terlipressin • 1 mg twice daily.		
Outcomes	- Duration of follow-up 15 days.		
Country of origin	China.		
Interventions	Terlipressin versus no intervention for 15 days.		
Proportion with type 1 hepatorenal syn- drome	Not reported.		
Notes	Full paper article.		
Risk of bias			
Bias	Authors' judgement	Support for judgement	
Random sequence generation (selection bias)	Unclear risk	Not described.	
Allocation concealment (selection bias)	Unclear risk	Not described.	

## Yang 2001 (Continued)

Blinding (performance bias and detection bias) All outcomes	High risk	Open trial.
Incomplete outcome data (attrition bias) All outcomes	Unclear risk	Unclear reporting of losses to follow-up.
Selective reporting (reporting bias)	Low risk	Clinically relevant outcome measures reported.
Other bias	Unclear risk	Sample size calculations are not reported.

SEM = standard error of mean.

g = gram.

## Characteristics of excluded studies [ordered by study ID]

Study	Reason for exclusion
Alessandria 2007	Randomised trial on noradrenaline plus albumin versus terlipressin plus albumin for hepatorenal syndrome
Angeli 2008	Randomised trial comparing different modes of administering terlipressin plus albumin for hepatorenal syndrome
Cavalli 2012	Randomised trial on terlipressin and albumin versus midodrine and octreotide and albumin for hepatorenal syndrome
Chelarescu 2003	Randomised trial comparing captopril plus octreotide versus octreotide published in abstract form
Pomier 2003	Cross-over trial on octreotide for hepatorenal syndrome.
Sharma 2008	Randomised trial on noradrenalin plus albumin versus terlipressin plus albumin for hepatorenal syndrome
Silawat 2011	Randomised trial on dopamine versus terlipressin plus albumin for hepatorenal syndrome

## Characteristics of ongoing studies [ordered by study ID]

## NCT00742339

Trial name or title	Terlipressin + albumin versus midodrine + octreotide in the treatment of hepatorenal syndrome
Methods	Randomised trial.
Participants	Patients with type 1 or 2 hepatorenal syndrome.

Terlipressin for hepatorenal syndrome (Review)

## NCT00742339 (Continued)

Interventions	Terlipressin plus albumin versus midodrine plus octreotide.
Outcomes	The primary outcome measure will be normalisation of creatinine at the end of treatment
Starting date	May 2005.
Contact information	Dr Angeli +390498218676 pangeli@unipd.it
Notes	Estimated enrolment 100 patients.

## NCT01143246

Trial name or title	A Placebo-Controlled, Double-Blind Study to Confirm the Reversal of Hepatorenal Syndrome Type 1 With Terlipressin
Methods	Study type: interventional. Study design: allocation: randomised. Endpoint classification: safety/efficacy study. Intervention model: parallel assignment. Masking: double blind (subject, investigator). Primary purpose: treatment.
Participants	Cirrhosis, ascites and hepatorenal syndrome type 1.
Interventions	Drug: terlipressin. Blinded terlipressin reconstituted with 5 mL of sterile 0.9% sodium chloride solution for injection will be administered intravenously as a slow bolus injection over 2 minutes at a dose of 1 mg (1 vial) every 6 hours (4 mg/day). Other Name: Lucassin®. Drug: placebo. Lyophilized mannitol reconstituted with 5 mL of sterile 0.9% sodium chloride solution administered intra- venously as a slow bolus injection over 2 minutes at a dose of 1 mg (1 vial) every 6 hours (4 mg/day)
Outcomes	Confirmed HRS Reversal: the percentage of subjects with two serum creatinine (SCr) values of $\leq$ 133 µmol/ L (1.5 mg/dL) at least 48 hours apart, on treatment, and without intervening RRT or liver transplant
Starting date	September 2010.
Contact information	Diane Stebbins diane.stebbins@ikaria.com
Notes	Estimated enrolment:180

## DATA AND ANALYSES

Outcome or subgroup title	No. of studies	No. of participants	Statistical method	Effect size
1 Mortality	5	249	Risk Ratio (M-H, Random, 95% CI)	0.75 [0.59, 0.97]
2 Reversal of hepatorenal syndrome	4	234	Risk Ratio (M-H, Random, 95% CI)	3.76 [2.21, 6.39]
3 Improved renal function	4	234	Risk Ratio (M-H, Random, 95% CI)	2.00 [1.11, 3.62]
4 Adverse events	4		Risk Ratio (M-H, Random, 95% CI)	Subtotals only
4.1 Abdominal pain or diarrhoea	2	76	Risk Ratio (M-H, Random, 95% CI)	5.42 [0.66, 44.71]
4.2 Bacterial infection	1	46	Risk Ratio (M-H, Random, 95% CI)	0.75 [0.39, 1.43]
4.3 Cardovascular adverse events	4	234	Risk Ratio (M-H, Random, 95% CI)	7.26 [1.70, 31.05]
4.4 Chest pain	1	52	Risk Ratio (M-H, Random, 95% CI)	5.0 [0.25, 99.34]
4.5 Circulatory overload	1	46	Risk Ratio (M-H, Random, 95% CI)	1.75 [0.59, 5.17]
4.6 Gastrointestinal bleeding	1	46	Risk Ratio (M-H, Random, 95% CI)	0.67 [0.22, 2.05]
4.7 Hepatic encephalopathy	1	46	Risk Ratio (M-H, Random, 95% CI)	1.0 [0.68, 1.47]
4.8 Livedo reticularis	1	112	Risk Ratio (M-H, Random, 95% CI)	3.00 [0.12, 72.10]
4.9 Respiratory distress or acidosis	1	112	Risk Ratio (M-H, Random, 95% CI)	7.0 [0.37, 132.46]
5 Mortality in trials on type 1 hepatorenal syndrome	3	188	Risk Ratio (M-H, Random, 95% CI)	0.71 [0.51, 0.98]
6 Mortality in trials on terlipressin and albumin	4	234	Risk Ratio (M-H, Random, 95% CI)	0.77 [0.61, 0.97]
7 Mortality in trials with a low risk of selection bias	4	234	Risk Ratio (M-H, Random, 95% CI)	0.77 [0.61, 0.97]
8 Mortality analysed using a fixed effect model	5	249	Risk Ratio (M-H, Fixed, 95% CI)	0.76 [0.63, 0.91]

## Comparison 1. Terlipressin alone or with albumin versus no intervention or albumin

Terlipressin for hepatorenal syndrome (Review)

# Analysis I.I. Comparison I Terlipressin alone or with albumin versus no intervention or albumin, Outcome I Mortality.

Review: Terlipressin for hepatorenal syndrome

Comparison: I Terlipressin alone or with albumin versus no intervention or albumin

Outcome: I Mortality

Study or subgroup	Treatment	Control	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	H,Random,95% Cl		H,Random,95% Cl_
Solanki 2003	7/12	12/12	-=-	18.5 %	0.60 [ 0.37, 0.97 ]
Sanyal 2008	32/56	35/56	+	30.6 %	0.91 [ 0.67, 1.24 ]
Neri 2008	12/26	21/26	-	19.7 %	0.57 [ 0.36, 0.90 ]
Mart n-Llah 2008	17/23	19/23	+	30.4 %	0.89 [ 0.66, 1.22 ]
Yang 2001	0/8	3/7		0.8 %	0.13[0.01, 2.10]
<b>Total (95% CI)</b> Total events: 68 (Treatment) Heterogeneity: Tau <sup>2</sup> = 0.03; Test for overall effect: $Z = 2$ . Test for subgroup differences	<b>125</b> , 90 (Control) Chi <sup>2</sup> = 6.56, df = 4 (P .19 (P = 0.028) s: Not applicable	<b>124</b> = 0.16); 1 <sup>2</sup> =39%	•	100.0 %	0.75 [ 0.59, 0.97 ]
			0.005 0.1 1 10	200	
			Favours treatment Favours	control	

Terlipressin for hepatorenal syndrome (Review)

## Analysis 1.2. Comparison I Terlipressin alone or with albumin versus no intervention or albumin, Outcome 2 Reversal of hepatorenal syndrome.

Review: Terlipressin for hepatorenal syndrome

Comparison: I Terlipressin alone or with albumin versus no intervention or albumin

Outcome: 2 Reversal of hepatorenal syndrome

Study or subgroup	Treatment	Control		Risk Ratio M-	Weight	Risk Ratio
	n/N	n/N	H,Ka	ndom,95% Cl		H,Kandom,95% Cl
Mart n-Llah 2008	9/23	1/23			7.2 %	9.00 [ 1.24, 65.41 ]
Neri 2008	21/26	5/26		-	43.1 %	4.20 [ 1.87, 9.44 ]
Sanyal 2008	19/56	7/56		-	46.1 %	2.71 [ 1.24, 5.94 ]
Solanki 2003	5/12	0/12			3.6 %	.00 [ 0.67,  79.29 ]
Total (95% CI)	117	117		•	100.0 %	3.76 [ 2.21, 6.39 ]
Total events: 54 (Treatment),	I 3 (Control)					
Heterogeneity: $Tau^2 = 0.0$ ; C	$2hi^2 = 2.12$ , $df = 3$ (P =	0.55); I <sup>2</sup> =0.0%				
Test for overall effect: $Z = 4$ .	88 (P < 0.00001)					
Test for subgroup differences	:: Not applicable					
			0.005 0.1	1 10 200		
			Favours control	Favours treatment	t	

Terlipressin for hepatorenal syndrome (Review)

## Analysis I.3. Comparison I Terlipressin alone or with albumin versus no intervention or albumin, Outcome 3 Improved renal function.

Review: Terlipressin for hepatorenal syndrome

Comparison: I Terlipressin alone or with albumin versus no intervention or albumin

Outcome: 3 Improved renal function

Study or subgroup	Treatment	Control	R	isk Ratio M-	Weight	Risk Ratio M-
	n/N	n/N	H,Kan	dom,95% Cl		H,Kandom,95% Cl
Mart n-Llah 2008	10/23	2/23			13.6 %	5.00 [ 1.23, 20.35 ]
Neri 2008	25/26	16/26		•	50.0 %	1.56 [ 1.14, 2.14 ]
Sanyal 2008	16/56	10/56	-	-	32.2 %	1.60 [ 0.80, 3.22 ]
Solanki 2003	5/12	0/12	-		4.2 %	.00 [ 0.67,  79.29 ]
Total (95% CI)	117	117		<b>*</b>	100.0 %	2.00 [ 1.11, 3.62 ]
Total events: 56 (Treatment),	28 (Control)					
Heterogeneity: $Tau^2 = 0.16$ ;	$Chi^2 = 5.69, df = 3 (P + 1)$	= 0.13); 1 <sup>2</sup> =47%				
Test for overall effect: $Z = 2.2$	30 (P = 0.022)					
Test for subgroup differences	: Not applicable					
			0.005 0.1 1	10 200		
			Favours control	Favours treatment		

Terlipressin for hepatorenal syndrome (Review)

# Analysis I.4. Comparison I Terlipressin alone or with albumin versus no intervention or albumin, Outcome 4 Adverse events.

Review: Terlipressin for hepatorenal syndrome

Comparison: I Terlipressin alone or with albumin versus no intervention or albumin

Outcome: 4 Adverse events

Study or subgroup	Treatment	Control	Risk Ratio	Weight	Risk Ratio
	n/N	n/N	H,Random,95% Cl		H,Random,95% Cl
I Abdominal pain or diarrhoe	ea				
Neri 2008	4/26	0/26		53.9 %	9.00 [ 0.51, 159.15 ]
Solanki 2003	1/12	0/12		46.1 %	3.00 [ 0.13, 67.06 ]
Subtotal (95% CI)	38	38		100.0 %	5.42 [ 0.66, 44.71 ]
Total events: 5 (Treatment), 0	) (Control)				
Heterogeneity: Tau <sup>2</sup> = 0.0; C Test for overall effect: $Z = 1.5$	hi <sup>2</sup> = 0.27, df = 1 (P =	0.60); l <sup>2</sup> =0.0%			
2 Bacterial infection	57(1 - 0.12)				
Mart n-Llah 2008	9/23	12/23		100.0 %	0.75 [ 0.39, 1.43 ]
Subtotal (95% CI)	23	23	•	100.0 %	0.75 [ 0.39, 1.43 ]
Total events: 9 (Treatment), I	2 (Control)				
Heterogeneity: not applicable	20 (5 0.20)				
3 Cardovascular adverse ever	38 (P – 0.38) nts				
Mart n-Llah 2008	5/23	0/23		26.2 %	.00 [ 0.64,  88. 3 ]
Neri 2008	3/26	0/26		24.9 %	7.00 [ 0.38,  29.   ]
Sanyal 2008	3/56	0/56		24.4 %	7.00 [ 0.37,   32.46 ]
Solanki 2003	2/12	0/12		24.5 %	5.00 [ 0.27, 94.34 ]
Subtotal (95% CI)	117	117	•	100.0 %	7.26 [ 1.70, 31.05 ]
Total events: 13 (Treatment),	0 (Control)				
Heterogeneity: $Tau^2 = 0.0$ ; C	$hi^2 = 0.15, df = 3 (P =$	0.99); l <sup>2</sup> =0.0%			
Test for overall effect: $Z = 2.6$	67 (P = 0.0075)				
Neri 2008	2/26	0/26	——————————————————————————————————————	100.0 %	5.00 [ 0.25, 99.34 ]
Subtotal (95% CI)	26	26		100.0 %	5.00 [ 0.25, 99.34 ]
Total events: 2 (Treatment), 0	) (Control)				
Heterogeneity: not applicable	2				
Test for overall effect: $Z = 1.0$	06 (P = 0.29)				
Mart p Llab 2008	7/22	4/23		100.0 %	
	1125	7725		100.0 %	1.75 [ 0.57, 5.17 ]
		0.0	002 0.1 10 500		
		Favours	s experimental Favours control		(Continued)
					(continued)

Terlipressin for hepatorenal syndrome (Review)

Study or subgroup	Treatment	Control	Risk Ratio	Weight	( Continued) Risk Batio
	n/N	n/N	H,Random,95%	, voigne	HJAR I HAD H,Random,95%
Subtotal (95% CI)	23	23	↓	100.0 %	1.75 [ 0.59, 5.17 ]
Total events: 7 (Treatment), 4 Heterogeneity: not applicable Test for overall effect: $Z = 1.0$ 6 Gastrointestinal bleeding	4 (Control) 2 DI (P = 0.31)				
Mart n-Llah 2008	4/23	6/23		100.0 %	0.67 [ 0.22, 2.05 ]
<b>Subtotal (95% CI)</b> Total events: 4 (Treatment), 6 Heterogeneity: not applicable Test for overall effect: Z = 0.7 7 Hepatic encephalopathy	<b>23</b> 5 (Control) 2 71 (P = 0.48)	23	•	100.0 %	0.67 [ 0.22, 2.05 ]
Mart n-Llah 2008	16/23	16/23	-	100.0 %	1.00 [ 0.68, 1.47 ]
Subtotal (95% CI) Total events: 16 (Treatment), Heterogeneity: not applicable Test for overall effect: Z = 0.0 8 Livedo reticularis Sanval 2008	<b>23</b> 16 (Control) 2 0 (P = 1.0) 1/56	<b>23</b>	• 	100.0 %	<b>1.00 [ 0.68, 1.47 ]</b>
	<b>F</b> (	6/30 E.C		100.0 %	3.00 [ 0.12, 72.10 ]
<ul> <li>Total events: I (Treatment), C</li> <li>Heterogeneity: not applicable</li> <li>Test for overall effect: Z = 0.6</li> <li>9 Respiratory distress or acid</li> <li>Sanyal 2008</li> </ul>	90 (Control) 68 (P = 0.50) osis 3/56	0/56		100.0 %	7.00 [ 0.37, 132.46 ]
Subtotal (95% CI)	56	56		100.0 %	7 00 [ 0 37 132 46 ]
Total events: 3 (Treatment), C Heterogeneity: not applicable Test for overall effect: $Z = 1.2$	) (Control) 2 30 (P = 0.19)	90		100.0 /0	7.00 [ 0.57, 152.40 ]
			0.002 0.1 10 500		
			Favours experimental Favours control		

## Analysis 1.5. Comparison I Terlipressin alone or with albumin versus no intervention or albumin, Outcome 5 Mortality in trials on type I hepatorenal syndrome.

Review: Terlipressin for hepatorenal syndrome

Comparison: I Terlipressin alone or with albumin versus no intervention or albumin

Outcome: 5 Mortality in trials on type 1 hepatorenal syndrome

Study or subgroup	Treatment	Control		R	isk Ratio M-	Weight	Risk Ratio M-
	n/N	n/N		H,Ran	dom,95% Cl		H,Random,95% Cl
Sanyal 2008	32/56	35/56		-	ł	43.0 %	0.91 [ 0.67, 1.24 ]
Solanki 2003	7/12	12/12		-		27.7 %	0.60 [ 0.37, 0.97 ]
Neri 2008	12/26	21/26		-		29.3 %	0.57 [ 0.36, 0.90 ]
Total (95% CI)	94	94		•		100.0 %	0.71 [ 0.51, 0.98 ]
Total events: 51 (Treatme	nt), 68 (Control)						
Heterogeneity: $Tau^2 = 0.0$	04; Chi <sup>2</sup> = 3.84, df = 2	$(P = 0.15); I^2 = 489$	6				
Test for overall effect: Z =	= 2.08 (P = 0.038)						
Test for subgroup differen	ces: Not applicable						
			0.01	0.1 1	10 100	)	
			Favours t	reatment	Favours contro	l	

Terlipressin for hepatorenal syndrome (Review)

## Analysis 1.6. Comparison I Terlipressin alone or with albumin versus no intervention or albumin, Outcome 6 Mortality in trials on terlipressin and albumin.

Review: Terlipressin for hepatorenal syndrome

Comparison: I Terlipressin alone or with albumin versus no intervention or albumin

Outcome: 6 Mortality in trials on terlipressin and albumin

Study or subgroup	Treatment	Control		F	Risk Ratio M-		Weight	Risk Ratio M-
	n/N	n/N		H,Rar	idom,95% Cl			H,Random,95% Cl
Neri 2008	12/26	21/26					19.1 %	0.57 [ 0.36, 0.90 ]
Solanki 2003	7/12	12/12					17.8 %	0.60 [ 0.37, 0.97 ]
Sanyal 2008	32/56	35/56			-		31.7 %	0.91 [ 0.67, 1.24 ]
Mart n-Llah 2008	17/23	19/23			_		31.4 %	0.89 [ 0.66, 1.22 ]
Total (95% CI)	117	117		+			100.0 %	0.77 [ 0.61, 0.97 ]
Total events: 68 (Treatment), 87 (Control)								
Heterogeneity: Tau <sup>2</sup> = 0.02; Chi <sup>2</sup> = 4.75, df = 3 (P = 0.19); l <sup>2</sup> =37%								
Test for overall effect: Z = 2	.18 (P = 0.029)							
Test for subgroup difference	s: Not applicable							
			0.2	0.5	1 2	5		
			Favours t	reatment	Favours	control		

Terlipressin for hepatorenal syndrome (Review)

## Analysis 1.7. Comparison 1 Terlipressin alone or with albumin versus no intervention or albumin, Outcome 7 Mortality in trials with a low risk of selection bias.

Review: Terlipressin for hepatorenal syndrome

Comparison: I Terlipressin alone or with albumin versus no intervention or albumin

Outcome: 7 Mortality in trials with a low risk of selection bias

Study or subgroup	Treatment	Control		F	Risk Ratio M-		Weight	Risk Ratio M-
	n/N	n/N		H,Ran	dom,95% Cl			H,Random,95% Cl
Mart n-Llah 2008	17/23	19/23			_		31.4 %	0.89 [ 0.66, 1.22 ]
Neri 2008	12/26	21/26					19.1 %	0.57 [ 0.36, 0.90 ]
Solanki 2003	7/12	12/12					17.8 %	0.60 [ 0.37, 0.97 ]
Sanyal 2008	32/56	35/56			_		31.7 %	0.91 [ 0.67, 1.24 ]
Total (95% CI)	117	117		+			100.0 %	0.77 [ 0.61, 0.97 ]
Total events: 68 (Treatment),	87 (Control)							
Heterogeneity: Tau <sup>2</sup> = 0.02; (	Chi <sup>2</sup> = 4.75, df = 3 (P	= 0.19); l <sup>2</sup> =37%						
Test for overall effect: $Z = 2$ .	18 (P = 0.029)							
Test for subgroup differences	: Not applicable							
			0.2	0.5	2	5		
			Favours t	reatment	Favours	control		

Terlipressin for hepatorenal syndrome (Review)

## Analysis 1.8. Comparison 1 Terlipressin alone or with albumin versus no intervention or albumin, Outcome 8 Mortality analysed using a fixed effect model.

Review: Terlipressin for hepatorenal syndrome

Comparison: I Terlipressin alone or with albumin versus no intervention or albumin

Outcome: 8 Mortality analysed using a fixed effect model

Study or subgroup	Treatment	Control		Risk Ratio		Weight	Risk Ratio
	n/N	n/N		M-H,Fixed,95% CI			M-H,Fixed,95% CI
Sanyal 2008	32/56	35/56		-		38.4 %	0.91 [ 0.67, 1.24 ]
Solanki 2003	7/12	12/12		-		13.7 %	0.60 [ 0.37, 0.97 ]
Mart n-Llah 2008	17/23	19/23		-		20.8 %	0.89 [ 0.66, 1.22 ]
Neri 2008	12/26	21/26		-		23.0 %	0.57 [ 0.36, 0.90 ]
Yang 2001	0/8	3/7	_			4.1 %	0.13[0.01, 2.10]
Total (95% CI)	125	124		•		100.0 %	0.76 [ 0.63, 0.91 ]
Total events: 68 (Treatment),	90 (Control)						
Heterogeneity: $Chi^2 = 6.56$ ,	df = 4 (P = 0.16); $ ^2 = 3$	19%					
Test for overall effect: $Z = 2$ .	92 (P = 0.0035)						
Test for subgroup differences	: Not applicable						
			0.005	0.1 1 10	200		

Favours treatment Favours control

## APPENDICES

## **Appendix I. Searches**

Database	Dates searched	Search terms	References identified
Cochrane Hepato- Biliary Group Controlled Trials Register	January 2012	(terlipressin* OR glypressin* OR vasoconstric*) AND 'hepatorenal syndrom*'	26
MEDLINE (Ovid SP)	1946 to January 2012	<ol> <li>exp Vasoconstrictor Agents/</li> <li>(terlipressin* or glypressin* or vasoconstric*).mp. [mp=title, original title, abstract, name of substance word, subject heading word]</li> </ol>	44

Terlipressin for hepatorenal syndrome (Review)

## (Continued)

		<ul> <li>3. 1 or 2</li> <li>4. exp Hepatorenal Syndrome/</li> <li>5. hepatorenal syndrom*.mp. [mp= title, original title, abstract, name of substance word, subject heading word]</li> <li>6. 4 or 5</li> <li>7. 6 and 3</li> <li>8. (random* or blind* or placebo* or meta-analysis).mp. [mp=title, origi- nal title, abstract, name of substance word, subject heading word]</li> <li>9. 8 and 7</li> </ul>	
Cochrane Central Register of Controlled Trials (CENTRAL) in <i>The Cochrane Library</i>	Issue 1 of 12, 2012	<ul> <li>#1 MeSH descriptor Vasoconstrictor Agents explode all trees 8487</li> <li>#2 terlipressin* OR glypressin* OR vasoconstric* 3302</li> <li>#3 (#1 OR #2) 10371</li> <li>#4 MeSH descriptor Hepatorenal Syndrome explode all trees 37</li> <li>#5 hepatorenal syndrom* 120</li> <li>#6 (#4 OR #5) 120</li> <li>#7 (#3 AND #4) 23</li> </ul>	16
EMBASE (Ovid SP)	1974 to January 2012	<ol> <li>exp Terlipressin/</li> <li>exp Vasoconstrictor Agent/</li> <li>(terlipressin* or glypressin* or vasoconstric*).mp. [mp=title, abstract, subject headings, heading word, drug trade name, original title, device manufacturer, drug manufacturer name]</li> <li>1 or 3 or 2</li> <li>exp Hepatorenal Syndrome/</li> <li>hepatorenal syndrom*.mp. [mp=title, abstract, subject headings, heading word, drug trade name, original title, device manufacturer, drug manufacturer name]</li> <li>6 or 5</li> <li>4 and 7</li> <li>(random* or blind* or placebo* or meta-analysis).mp. [mp=title, abstract, subject headings word, drug trade name, original title, device manufacturer, drug manufacturer, abstract, subject headings word, drug trade name]</li> </ol>	160

(Continued)

Science Citation Index Ex- panded (http://apps.isiknowl- edge.com)	1900 to January 2012	# 5 96 #4 AND #3 # 4 999,592 TS=(random* or blind* or placebo* or meta-analysis) # 3 350 #1 AND #2 # 2 1,675 TS=(hepatorenal syn- drom*) # 1 34,324 TS=(terlipressin* or gly- pressin* or vasoconstric*)	96
--	----------------------	---	----

## WHAT'S NEW

Last assessed as up-to-date: 31 January 2012.

Date	Event	Description
9 April 2012	Amended	Based on peer review comments, exclusion of all other intervention comparisons than terlipressin alone or with albumin versus no intervention, placebo or albumin
9 March 2012	Amended	Based on peer review comments, the outcomes are changed with inclusion of quality of life according to the Cochrane Hepato-Biliary Group recommendations
11 January 2012	New search has been performed	New searches are performed, and additional trials were identified and included in the review
9 March 2011	Amended	Based on peer review comments, we changed the defi- nitions of primary outcomes. In the present version, all outcomes are defined as primary based on recommen- dations from peer review comments
11 January 2011	New citation required but conclusions have not changed	Additional trials are included (the original review in- cluded data from three trials and the updated review has data from eight trials). The overall conclusions are not changed

## CONTRIBUTIONS OF AUTHORS

LG drafted the original and updated review. Three authors LG, KC and AK have participated in the literature searches and data extraction. All authors have participated in revision of the review.

## DECLARATIONS OF INTEREST

None of the authors have financial, academic, or personal conflicts of interests with regard to the submitted work.

## SOURCES OF SUPPORT

#### Internal sources

• No internal funding received, Not specified.

#### **External sources**

• No external funding received, Not specified.

## DIFFERENCES BETWEEN PROTOCOL AND REVIEW

• The previous version of the present review included three trials (Hadengue 1998; Yang 2001; Solanki 2003). Based on the changes in the outcome measures assessed, we were unable to extract data from one of these trials (Hadengue 1998). We were therefore unable to include the trial in our analyses. The updated review includes six trials, but only includes data from five trials. Inclusion of the new trials confirmed the findings of the previous review.

• We have updated the criteria for assessment of the outcome measure on hepatorenal syndrome based on the latest criteria (described by the International Ascites Club (www.icascites.org/)). Based on these criteria, we have excluded urine output and creatinine clearance from our analyses.

• The assessment of bias control is revised, and the statistical analyses revised the included incorporation of trial sequential analysis.

• Following review comments and the guidelines specified by the Cochrane Hepato-biliary Group, we added quality of life as an outcome measure and classed all of our outcome measures as primary outcomes.

• We originally planned to include trials comparing different vasoactive drugs, but excluded these analyses based on review comments.

## INDEX TERMS

## Medical Subject Headings (MeSH)

Albumins [therapeutic use]; Hepatorenal Syndrome [classification; \*drug therapy]; Lypressin [\*analogs & derivatives; therapeutic use]; Randomized Controlled Trials as Topic; Vasoconstrictor Agents [\*therapeutic use]

### MeSH check words

Humans